Author's response to reviews

Title: Sexual behaviour, Contraceptive Knowledge and Use among Female University Students in Dar es Salaam, Tanzania: a cross-sectional study

Authors:

Michael J Mahande MJM (jmmahande@gmail.com)
Magreat J Somba MJS (magreats@yahoo.com)
Milline Mbonile MM (mmbonile@udsm.ac.tz)
Joseph Obure JO (jobure2002us@yahoo.com)

Version: 2 Date: 18 December 2013

Author's response to reviews: see over
Author's response to editorial comments

Title: Sexual behaviour, Contraceptive Knowledge and Use among Female University Students in Dar es Salaam, Tanzania: a cross-sectional study

Authors:

MJM: jmmahande@gmail.com
MM: mmbonile@udsm.ac.tz
MJS: magreats@yahoo.com
JO: jobure2002us@yahoo.com

Version: 1 Date: 18th December 2013

Dear Editor,

Thank you very much for the invitation to resubmit our manuscript entitled “Sexual behaviour, Contraceptive Knowledge and Use among Female University Students in Dar es Salaam, Tanzania: a cross-sectional study” and thanks for the opportunity to further improve the manuscript. All the editorial suggestions have been adopted, and we hope the paper now is acceptable for further review in your journal. Please find our responses indicated in a red colour.

Section Editor's Comment:
----------------------------------------

1. Competing Interests:

Please be advised that manuscripts must include a ‘Competing interests’ section. This should be placed after the Conclusions/Abbreviations. If there are none to declare, please include the statement “The authors declare that they have no competing interests.” Please consider the following questions and include an appropriate declaration of competing interests in your manuscript:
The declaration for competing interest has been addressed as per editorial suggestion and inserted after the conclusion. All authors declared that they have no competing interests either in terms of financial or non-financial competing interests.

2. **Title Page:**

Please include on the title page, at minimum, the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author.

We have now included names of all authors with their respective institutions and country of residence for each author. In addition, email addresses of all authors and full address of the submitting author are available on the title page of the manuscript.

---------------------------------